Author's response to reviews

Title: Experiences of undergoing cardiac surgery among older people diagnosed with postoperative delirium: one year follow-up

Authors:

Helena Claesson Lingehall (helena.lingehall@umu.se)
Nina Smulter (nina.smulter@umu.se)
Birgitta Olofsson (birgitta.olofsson@umu.se)
Elisabeth Lindahl (elisabeth.lindahl@umu.se)

Version: 2 Date: 2 January 2015

Author's response to reviews: see over
Dear Editor

First, we appreciate your encouragement to submit our manuscript once more title “Experiences of undergoing cardiac surgery among older people diagnosed with postoperative delirium: one year follow-up”.

The point-by-point descriptions of the changes are made:

I have followed your abstract headings. The manuscript now includes line and page numbers.

Sincerely

Helena Claesson Lingehall

Address for correspondence:
Helena Claesson Lingehall, RN, MSc, PhD-student, Department of Nursing, Umeå, Sweden, Umeå University
SE- 901 87 Umeå SWEDEN
helena.lingehall@umu.se
telephone home +46 90191418
telephone office +46 0700402666
fax number +46 907867508